Author's response to reviews

Title: Expression and prognostic significance of THBS1, Cyr61 and CTGF in esophageal squamous cell carcinoma

Authors:

zhuqing zhou (s_zqzhou@163.com)
weihua cao (zhw1405@126.com)
jianjun xie (g_jixie@stu.edu.cn)
jing lin (linjingst73@hotmail.com)
zhongying shen (zhongyingshen@yahoo.com)
qingying zhang (qyzhang@stu.edu.cn)
jinhui shen (enmin2007@yahoo.com)
liyan xu (liyanxu1130@yahoo.com.cn)
enmin li (nmli@stu.edu.cn)

Version: 3 Date: 17 August 2009

Author's response to reviews: see over
Dear Editor:

We are now submitting the revised version of our manuscript entitled *Expression and prognostic significance of THBS1, Cyr61 and CTGF in esophageal squamous cell carcinoma* for publication in BMC Cancer (MS: 6382342562458000).

According to the comments of the editor, we have made the following revision:

1. Remove all the additional files from the system.
2. Used sentence case in the title both in the manuscript files and the submission system.
3. Renamed Material and methods as Methods.

All authors have read and approve this revised version of the manuscript. No part of this paper has been published or submitted elsewhere. No conflict of interest exists for any of the authors.

Sincerely,

En-Min Li, Ph.D.
Professor
Department of Biochemistry and Molecular Biology
Medical College, Shantou University
Guangdong Province
Shantou, 515041
China
Tel.: 86-754-88900847
Fax: 86-754-88900847
E-mail: nmli@stu.edu.cn